Author's response to reviews

Title: Cold Saline Irrigation of the Renal Pelvis During Radiofrequency Ablation of a Central Renal Neoplasm

Authors:

Jose Pablo Morales (jpmorales@doctors.org.uk)
Tarun Sabharwal (Tarun.Sabharwal@gstt.nhs.uk)
Marios Georganas (gmarios@doctors.org.uk)
Renato Dourado (Renato.Dourado@gstt.nhs.uk)
Declan Cahill (Declan.Cahill@gstt.nhs.uk)
Andreas Adam (Andy.Adam@kcl.ac.uk)

Version: 5 Date: 11 June 2007

Author's response to reviews: see over
11 June 2007

Professor Michael Kidd

Editor-in-Chief
Journal of Medical Case Reports
Professor and Head of the Discipline of General Practice
The University of Sydney

Dear Editor,


Authors: JP Morales, T Sabharwal, M Georganas, R Dourado, D Cahill, A Adam

Thank you very much for considering our paper for publication in the Journal of Medical Case Reports. We have responded to the reviewers’ comments and made changes as requested. Please find enclosed the edited version of our manuscript, figures and the response letter to reviewers’.

There are no potential or actual competing interests.

Yours sincerely,

The Authors